Author's response to reviews

Title: Individual patient data meta-analysis of diagnostic and prognostic studies in obstetrics, gynaecology and reproductive medicine.

Authors:

Kimiko A Broeze (k.a.broeze@amc.uva.nl)
Brent C Opmeer (b.c.opmeer@amc.uva.nl)
Lucas M Bachmann (lucas.bachmann@usz.ch)
Frank J Broekmans (F.Broekmans@umcutrecht.nl)
Patrick MM Bossuyt (p.m.bossuyt@amc.uva.nl)
Sjors FPJ Coppus (S.F.Coppus@amc.uva.nl)
Neil P Johnson (n.johnson@auckland.ac.nz)
Khalid S Khan (k.s.khan@bham.ac.uk)
Gerben Ter Riet (G.terRiet@amc.uva.nl)
Fulco Van der Veen (f.vanderveen@amc.uva.nl)
Madelon Van Wely (m.vanwely@amc.uva.nl)
Ben WJ Mol (b.w.mol@amc.uva.nl)

Version: 5 Date: 16 March 2009

Author's response to reviews:

Dear editor,

We changed formats as requested in your mail. In the last previous submitted versions of our manuscript we forgot to mention Sjors Coppus as a co-author, so we added him in the list of co-authors as well in the manuscript. We hope this is not a problem for you. In stead of one figure we now submitted figure 1 as four separate figures.

On behalf of the other authors, yours sincerely,

Kimiko Broeze